Author’s response to reviews

Title: Stability of glycoprotein gene sequences of herpes simplex virus type 2 from primary to recurrent human infection, and diversity of the sequences among patients attending an STD clinic.

Authors:

Lars Haarr (lars.haarr@k2.uib.no)
Arvid Nilsen (arvid.nilsen@helse-bergen.no)
Per M. Knappskog (per.knappskog@helse-bergen.no)
Nina Langeland (Nina.Langeland@mofa.uib.no)

Version: 2 Date: 21 September 2013

Author’s response to reviews:

We were asked to include ethics and consent statement in the Methods section. We apologize for not having done so in the first version of the manuscript. This has now been corrected, and we hope that what is added is acceptable.